Author's response to reviews

Title: Hornerin, an S100 family protein, is functional in breast cells and aberrantly expressed in breast cancer

Authors:

Jodie M Fleming (jodie.fleming@nccu.edu)
Erika Ginsburg (ginsbure@mail.nih.gov)
Shannon Oliver (olivers@nccu.edu)
Paul Goldsmith (paulg@mail.nih.gov)
Barbara K Vonderhaar (vonderhb@mail.nih.gov)

Version: 2 Date: 5 March 2012

Author's response to reviews: see over
Dear Editor:

Please find, attached, our revised manuscript entitled “Hornerin, an S100 family protein, is functional in breast cells and aberrantly expressed in breast cancer” by Fleming et al.

We have edited the manuscript to include the requested information. The manuscript now includes the following: “Collection of patient samples was performed in accordance with the Helsinki Declaration under the guidelines of the National Cancer Institute Review Board, protocol number 99-CC-0168. Written informed consent was obtained from all human subjects as specified in the protocol.”

We hope you will find our manuscript suitable for publication in BMC Cancer. The manuscript contains original data that are not under consideration for publication elsewhere. All authors declare that they have no competing interests.

Sincerely,

Jodie M. Fleming